Author's response to reviews

Title: Pituitary metastasis of rhabdomyosarcoma: First case and literature revue.

Authors:

ESSADI E.I Ismail (ismail_onco@yahoo.fr)
SBITTI S.Y Yassir (sbittiyassir@gmail.com)
LALYA L.I Issam (lalyaissam@yahoo.fr)

Version: 2 Date: 19 December 2013

Author's response to reviews: see over
Pituitary metastasis of rhabdomyosarcoma: First case and literature review

Essadi Ismail (a); Yassir Sbitti (b); Lalya Issam (c)

(a) Ibn Sina Military Hospital, Medical Oncology, Marrakesh, Morocco
(b) Mohamed V Military Hospital, Medical Oncology, Rabat, Morocco
(c) Mohamed V Military Hospital, Radiation Oncology, Rabat, Morocco

Corresponding author: Essadi Ismail

Mail: ismail_onco@yahoo.fr

Cover letter

To the editorial board of the Journal of Medical Case Reports

Dear Editor-in-Chief,

I am sending you our manuscript entitled "Pituitary metastasis of rhabdomyosarcoma: First case and literature review" by Essadi I. et al. We would like to have the manuscript considered for publication, because we believe this is the first case of its kind ever published in the literature. Publication of this article will educate the medical community about the possibility of the occurrence of pituitary metastases in rhabdomyosarcoma, which must actively seek to have access to a suitable therapeutic approach. There is no conflict of interest.

Please let me know of your decision at your earliest convenience.

With my best regards,

Sincerely yours,

ESSADI I., MD